Author's response to reviews

Title: Necrotic gangrenous intrathoracic appendix in a marfanoid adult patient. A case report

Authors:

Mohannad J Barakat (barakat@blueyonder.co.uk)
Jon H Vickers (jvickers@waht.swest.nhs.uk)

Version: 2 Date: 11 December 2004

Author's response to reviews:

1) The background has been moved to the discussion and the discussion has been expanded
2) Yes the patient did exhibit specific features of Marfan's syndrome and did have a positive family history, but had never undergone family testing and had never thought about it as she had been well till her admission
3) There is three publications associating Marfan's syndrome and diaphragmatic hernias in my report.